Author's response to reviews

Title: Relationship between EGFR expression, copy number and mutation in lung adenocarcinomas

Authors:

Zhiyong Liang (liangzhiyong1220@yahoo.com)
Jing Zhang (jingzhangsmile@163.com)
Xuan Zeng (zengx88@yahoo.com.cn)
Jie Gao (zy_795@hotmail.com)
Shafei Wu (Suphie@sina.com)
Tonghua Liu (liuth_pumch@yahoo.com.cn)

Version: 6 Date: 16 July 2010

Author's response to reviews:

Some typing errors in discussion part (‘denocarcinoma’ was changed to 'adenocarcinoma') and author's contribution part ('X Z participated' was changed to 'XZ participated', 'S W carried' was changed to 'SW carried') were changed.